Author's response to reviews

Title: Investigating the health implications of social policy initiatives at the local level: study design and methods

Authors:

Gemma E Carey (gcarey@unimelb.edu.au)

Version: 4 Date: 8 December 2010

Author's response to reviews:

Dear Editorial Team,

Thank you for accepting the manuscript in principle. The following formatting changes have been made:

1. The author details have been amended
2. Referencing formatting has been changed.
3. Title page has been removed
4. Affiliation now includes city and country.
5. Track changes has been removed.
6. Methods has been changed to 'Methods/Design'
7. Abbreviations have been reformatted.

I hope these changes are acceptable.

Sincerely,

Gemma Carey